Author's response to reviews

Title: High-Grade Symptomatic or Asymptomatic Carotid Stenosis in the Very Elderly: A Challenge for Proponents of Carotid Angioplasty and Stenting.

Authors:

Enzo Ballotta (enzo.ballotta@unipd.it)
Giuseppe Da Giau (stonetto@einfo.it)
Carmelo Militello (carmelo.militello@unipd.it)
Bruno Barbon (bruno.barbon@email.it)
Aldo De Rossi (derossialdo@libero.it)
Giorgio Meneghetti (giorgio.meneghetti@unipd.it)
Claudio Baracchini (cbaracchini@ulss.tv.it)

Version: 3 Date: 21 March 2006

Author's response to reviews: see over
Padova, March 21, 2006

Editor
BMC Cardiovascular Disorders.

Dear Editor:

Thank you for reviewing the revised version of our manuscript entitled: “High-Grade Symptomatic and Asymptomatic Carotid Stenosis in the Very Elderly: A Challenge for Proponents of Carotid Angioplasty and Stenting”.

Uploaded, please find the revised version 2 of our manuscript. The manuscript has been resubmitted according to your suggestions. A point-by-point response to each suggestion is provided below.

**Suggested Changes:**

- **Title**: Please remove the full stop at the end of the title.  OK
- **Title page**: Please remove the running title, the word counts and the article type from the title page.  OK
- **Keywords**: Please remove the keyword section.  OK
- **Tables**: Please format the tables in black and white.  OK

Thank you for the opportunity to revise and resubmit the paper.

We agree to proceed with the publication of the manuscript in *BMC Cardiovascular Disorders*.

Sincerely,

Enzo Ballotta, MD